For reprint orders, please contact: reprints@futuremedicine.com

Encorafenib plus binimetinib in patients with BRAF ^{V600}-mutant non-small cell lung cancer: phase II PHAROS study design

Gregory J Riely^{*,1}^(b), Myung-Ju Ahn²^(b), Enriqueta Felip³^(b), Suresh S Ramalingam⁴^(b), Egbert F Smit⁵^(b), Anne S Tsao⁶^(b), Ann Alcasid⁷, Tiziana Usari⁸, Paul S Wissel⁷^(b), Keith D Wilner⁹^(b) & Bruce E Johnson¹⁰

¹Department of Medicine, Memorial Sloan Kettering Cancer Center, New York, NY 10065, USA

²Department of Hematology and Oncology, Samsung Medical Center, Seoul, 06351, Republic of Korea

³Oncology Department, Vall d'Hebron Institute of Oncology, Vall d'Hebron University Hospital, Barcelona, 08035, Spain

⁴Department of Hematology and Medical Oncology, Emory University School of Medicine, Atlanta, GA 30322, USA

⁵Department of Thoracic Oncology, Netherlands Cancer Institute, Amsterdam, 1066, CX, The Netherlands

⁶Department of Thoracic/Head and Neck Medical Oncology, MD Anderson Cancer Center, TX 77030, Houston

⁷Clinical Development and Operations, Pfizer Inc., Collegeville, PA 19426, USA

⁸Biostatistics, Pfizer, Milan, 20152, Italy

⁹Clinical Development and Operations, Pfizer Inc., San Diego, CA 92121, USA

¹⁰Department of Medical Oncology, Dana-Farber Cancer Institute, Boston, MA 02215, USA

*Author for correspondence: rielyg@mskcc.org

BRAF^{V600} oncogenic driver mutations occur in 1–2% of non-small-cell lung cancers (NSCLCs) and have been shown to be a clinically relevant target. Preclinical/clinical evidence support the efficacy and safety of BRAF and MEK inhibitor combinations in patients with NSCLC with these mutations. We describe the design of PHAROS, an ongoing, open-label, single-arm, phase II trial evaluating the BRAF inhibitor encorafenib plus the MEK inhibitor binimetinib in patients with metastatic *BRAF* ^{V600}-mutant NSCLC, as first- or second-line treatment. The primary end point is objective response rate, based on independent radiologic review (per RECIST v1.1); secondary objectives evaluated additional efficacy end points and safety. Results from PHAROS will describe the antitumor activity/safety of encorafenib plus binimetinib in patients with metastatic *BRAF* ^{V600}-mutant NSCLC.

Plain language summary: Some people with non-small-cell lung cancer (NSCLC) have changes in a gene called *BRAF* (known as 'gene mutations'). One common *BRAF* mutation is called 'V600'. Combinations of medicines that block proteins encoded by mutant *BRAF* and another gene called *MEK* can shrink tumors and slow their progression. We describe the design of PHAROS, a clinical trial investigating encorafenib (mutant BRAF inhibitor) combined with binimetinib (MEK inhibitor) in people with *BRAF* ^{V600}-mutant NSCLC that had spread to other parts of the body ('metastatic disease'). People are monitored for side effects and to see if their tumor shrunk. PHAROS includes people treated with encorafenib plus binimetinib as their first treatment for metastatic disease, and people whose cancer progressed after previous anticancer therapy.

Clinical trial registration: Clinicaltrials.gov (NCT03915951) and EudraCT (2019-000417-37)

First draft submitted: 13 October 2021; Accepted for publication: 17 November 2021; Published online: 17 December 2021

Keywords: binimetinib • *BRAF*^{V600} mutation • clinical trial • efficacy • encorafenib • non-small-cell lung cancer • objective response rate • safety

Molecularly targeted therapies are increasingly being developed for biomarker-selected patients with non-smallcell lung cancer (NSCLC) characterized by specific oncogenic driver mutations and gene rearrangements [1,2]. Significant advances in targeted therapy have been associated with improvements in population-level mortality in NSCLC in recent years [3].



Future



Figure 1. BRAF and MEK inhibitor combination mechanism of action. (A) Activating *BRAF* mutations drive aberrant cell growth and proliferation through constitutive MAPK pathway activation. **(B)** The combination of encorafenib plus binimetinib targets two kinases within the MAPK pathway to inhibit deregulated growth and proliferation caused by *BRAF* driver mutations.

Activating *BRAF* driver mutations occur in approximately 3–4% of patients with NSCLC [4–11], with most of these mutations (~85%) being observed in adenocarcinomas [9–11]. While they are a small subset of all patients with NSCLC, some of the recent and significant advances in NSCLC treatment have been seen with use of novel therapies targeted to specific mutations [1,2]. Moreover, since NSCLC is quite prevalent (i.e., >2 million new cases and >1.7 million deaths annually worldwide, based on 2018 estimates [12]), patients with *BRAF*-mutant NSCLC represent a relatively large number of patients.

The most common activating *BRAF* mutations occur on codon 600 (*BRAF*^{V600}), with most cases involving the *BRAF*^{V600E} point mutation, which represents approximately 50% of all *BRAF* mutations in lung cancer [13]. There is variability across studies in the identified associations between *BRAF* mutations and clinical and pathological NSCLC features; however, *BRAF* mutations appear to occur more frequently in patients with adenocarcinoma, a history of smoking, and in women [7–10]. There is also evidence that *BRAF* ^{V600E}-mutated tumors are associated with an aggressive phenotype, poor prognosis and lack of chemosensitivity to platinum-based chemotherapy [7–9].

Rationale & clinical proof-of-concept validation for BRAF plus MEK inhibitor combinations

BRAF is part of the MAPK pathway, which controls cell growth and proliferation (Figure 1). Activating *BRAF* mutations act as oncogenic drivers by causing constitutive activation of downstream MAPK pathway signaling, resulting in unchecked cell growth and proliferation [9]. In targeting *BRAF*-mutant cancers, BRAF inhibitors are typically used in combination with inhibitors of the downstream kinase MEK. Targeting two kinases within the same RAS/RAF/MEK/ERK pathway achieves a greater antitumor activity and prolongs progression-free survival (PFS) [9,10,14,15]. The addition of a MEK inhibitor also mitigates the paradoxical activation of the MAPK pathway and associated adverse events (AEs) that can occur as a result of BRAF inhibitor monotherapy [14,16,17]. Multiple BRAF and MEK inhibitor combinations have been validated clinically in the treatment of *BRAF* ^{V600E/K}-mutant metastatic or unresectable melanoma where such combinations are an established standard of care [18–23].

In *BRAF* ^{V600E}-mutant NSCLC, combined BRAF/MEK inhibition is associated with better response rates and PFS compared with BRAF inhibitor monotherapy. In an open-label, phase II clinical trial, the combination of the BRAF inhibitor dabrafenib plus the MEK inhibitor trametinib elicited an objective response rate (ORR; primary end point), as determined by the investigator, of 64% (23/36 patients; 95% CI: 46–79) for treatment-naive patients and 63% (36/57 patients; 95% CI: 49–76) for patients who received prior systemic chemotherapy with *BRAF* ^{V600E}-mutant metastatic NSCLC [24,25]. In comparison, dabrafenib monotherapy was associated with

an ORR of 33% (95% CI: 23–45) for previously treated patients (four of six treatment-naive patients also responded) [26]. Among responders who received dabrafenib plus trametinib, the median duration of response (DOR) was 10.4 months (95% CI: 8.3–17.9) for treatment-naive patients and 9.0 months (95% CI: 6.9–18.3) for previously treated patients [24,25]. After a minimum of 5-year follow-up, the 4- and 5-year survival rates were 34 and 22%, respectively, in treatment-naive patients (median overall survival [OS]: 17.3 months) and 26 and 19%, respectively, in pretreated patients (median OS: 18.2 months) and patients who received dabrafenib plus trametinib [27]. Based on results from this trial, the combination of dabrafenib and trametinib is a current standard-of-care regimen for *BRAF* V^{600E}-mutant metastatic NSCLC in the first-line setting or after progression on first-line platinum-based therapy with or without a PD-1/PD-L1 inhibitor [1,28]. The National Comprehensive Cancer Network NSCLC panel also currently recommends that all patients with metastatic nonsquamous NSCLC be tested for *BRAF* mutations and recommends initiation of targeted therapies for those patients whose tumors harbor *BRAF*^{V600E} [28].

Encorafenib in combination with binimetinib

Encorafenib and binimetinib have been developed as selective BRAF and MEK inhibitors, respectively [29,30]. Encorafenib has a relatively long dissociation half-life from mutant $BRAF^{V600E}$ (>30 h) compared with other BRAF inhibitors (e.g., 2 h for dabrafenib and 0.5 h for vemurafenib), resulting in prolonged target inhibition and higher potency compared with BRAF inhibitors that dissociate more quickly [31,32]. The clinical efficacy of the combination of encorafenib plus binimetinib versus vemurafenib monotherapy in $BRAF^{V600E/K}$ -mutant melanoma was established in the COLUMBUS trial based on improved PFS and OS [22,33]. The combination of encorafenib plus binimetinies as a common treatment-limiting toxicity of substantial concern associated with dabrafenib plus trametinib, and as such pyrexia management algorithms are in development to reduce related adverse outcomes [35,36]. While pyrexia has been observed in more than half of patients treated with dabrafenib plus trametinib, pyrexia was relatively less frequent (18% of patients overall) in patients treated with encorafenib plus binimetinib in the COLUMBUS trial [17,35]. Most cases of pyrexia in COLUMBUS were mild (66%; no patients experienced grade 4 pyrexia) and led to dosing interruption/adjustments in 4% of patients and discontinuation in less than 1% of patients [17]. Based on the COLUMBUS trial, the combination of encorafenib plus binimetinib was approved for the treatment of patients with $BRAF^{V600E/K}$ -mutant unresectable or metastatic melanoma [37].

PHAROS trial

Given the manageable safety profile and antitumor activity in patients with melanoma, encorafenib plus binimetinib may represent an alternative BRAF/MEK inhibitor combination for the treatment of $BRAF^{V600}$ -mutant metastatic NSCLC.

Objectives

We report the design of the PHAROS trial (NCT03915951), which was initiated to explore the antitumor activity and safety of encorafenib plus binimetinib in patients with $BRAF^{V600}$ -mutant metastatic NSCLC, who are either treatment-naive or who have been previously treated with platinum-based chemotherapy and/or anti-PD-1/PD-L1 inhibitor therapy.

Study design

PHAROS is an ongoing, open-label, single-arm, phase II trial evaluating encorafenib plus binimetinib in patients with metastatic $BRAF^{V600}$ -mutant NSCLC. The overall study design is summarized in Figure 2. Eligible patients receive encorafenib 450 mg once daily (q.d) plus binimetinib 45 mg twice daily (b.i.d), administered orally. These dosages are consistent with the approved doses and administration schedule for this combination in patients with $BRAF^{V600E/K}$ -mutant unresectable or metastatic melanoma [37,38]. Treatment is administered until disease progression, unacceptable toxicity, withdrawal of consent, initiation of subsequent anticancer therapy, death or end of study.

Dose reductions and temporary dosing interruptions are permitted for toxicity management, and in cases of medical or surgical events or logistical reasons unrelated to the study. However, encorafenib and/or binimetinib will be permanently discontinued for patients missing more than 6 consecutive weeks of dosing with either or both study drug(s). Patients permanently discontinuing treatment with encorafenib must also discontinue treatment

Clinical Trial Protocol Riely, Ahn, Felip et al.



Figure 2. PHAROS trial design.

AE: Adverse event; b.i.d: Twice daily; BRAFi: BRAF inhibitor; DCR: Disease control rate; DOR: Duration of response; ECOG PS: Eastern Cooperative Oncology Group performance status; *EGFR*m: *EGFR* mutation; FNA: Fine needle aspiration; IRR: Independent radiology review; MEKi: MEK inhibitor; NGS: Next-generation sequencing; NSCLC: Non-small-cell lung cancer; ORR: Objective response rate; OS: Overall survival; PD: Progressive disease; PFS: Progression-free survival; q.d: Once daily.

with binimetinib. Those permanently discontinuing treatment with binimetinib may continue treatment with encorafenib, but encorafenib dose modification may be needed due to the potential for increased toxicity of encorafenib administered without binimetinib. Following a dose reduction, the dose can be re-escalated to the next higher dose level at the discretion of the investigator once the AE that resulted in dose reduction improves and remains stable to the patient's baseline for ≥ 14 days in the absence of concomitant toxicities. Dose re-escalations are not allowed after a dose reduction due to prolonged QTcF ≥ 501 ms (encorafenib), left ventricular ejection fraction dysfunction (binimetinib) or ocular toxicity grade ≥ 2 (both drugs).

The study is being performed in accordance with the requirements of the applicable local regulatory authorities and International Council of Harmonisation good clinical practice guidelines. Institutional review board/independent ethics committee approval of the protocol and all amendments was required prior to implementation.

Key eligibility criteria

As summarized in Table 1, the study includes adults (\geq 18 years) with histologically confirmed stage IV or metastatic NSCLC and measurable disease based on Response Evaluation Criteria in Solid Tumors (RECIST) v1.1 criteria. Two cohorts are being enrolled, a cohort including patients who have had no prior treatment for metastatic disease and another cohort who have received either first-line platinum-based chemotherapy or first-line anti-PD-1/PD-L1 inhibitor treatment (alone or in combination with another immunotherapy and/or platinum-based chemotherapy). Patients will be eligible based on identification of the *BRAF*^{V600E} mutation in tumor tissue or blood, as determined by a PCR or next-generation sequencing-based local laboratory assay. Other less common class 1 *BRAF*^{V600} mutations (e.g., K or D) are also permitted. While patients must have tumor tissue available for central laboratory confirmation of *BRAF*^{V600} mutations, central laboratory testing is not used to determine study eligibility. Patients must have written documentation from a local pathology report of a *BRAF*^{V600} mutation in tumor tissue or blood. Patients who have received prior treatment with a BRAF or MEK inhibitor are excluded. In addition, patients with other driver mutations (*EGFR* mutation, *ALK* rearrangement or *ROS1* rearrangement) or untreated symptomatic brain metastasis, leptomeningeal disease or other active central nervous system metastases are not eligible for the study. All the patients must provide written informed consent before enrollment.

Table 1. Key inclusion and exclusion criteria.		
Key inclusion criteria		
Demographics	• Aged \geq 18 years	
NSCLC disease characteristics	 Histologically confirmed NSCLC stage IV or metastatic disease (M1a, M1b, M1c – AJCC 8th edition) Presence of measurable disease (per RECIST v1.1) ECOG performance status 0 or 1 Presence of <i>BRAF</i>^{V600} mutation[†] Adequate tumor tissue for submission to central laboratory for confirmation of mutation status 	
Prior treatment	 Treatment-naive OR Prior treatment with: First-line platinum-based chemotherapy OR First-line treatment with an anti-PD-1/PD-L1 inhibitor alone or in combination with platinum-based chemotherapy, or in combination with immunotherapy with or without platinum-based chemotherapy 	
Key exclusion criteria		
NSCLC disease characteristics	Documented EGFR mutation, ALK fusion oncogene or ROS1 rearrangement Symptomatic brain metastasis, leptomeningeal disease or other active CNS metastases	
Prior treatment	Prior treatment with any BRAF inhibitor or MEK inhibitor	
Other	• Evidence of active noninfectious pneumonitis or history of interstitial lung disease	
[†] BRAF ^{V600E} : other less common C	lass 1 BRAF ^{V600} mutations (e.g., K or D) permitted with prior discussion with the Sponsor.	

AJCC: American Joint Committee on Cancer; ECOG: Eastern Cooperation Oncology Group; NSCLC: Non-small-cell lung cancer; RECIST: Response Evaluation Criteria in Solid Tumors.

Table 2. Study objectives and end points.			
Primary objective	Primary end point		
• To evaluate the antitumor activity of encorafenib plus binimetinib in treatment-naive and previously treated patients with <i>BRAF</i> ^{V600E} -mutant NSCLC as measured by ORR	Confirmed ORR as determined by IRR per RECIST v1.1 in treatment-naive and previously treated patients		
Secondary objectives	Secondary end points		
• To evaluate the antitumor activity of encorafenib plus binimetinib in treatment-naive and previously treated patients with <i>BRAF</i> ^{V600E} -mutant NSCLC as measured by ORR, DOR, TTR [†] , DCR, PFS and OS	 Confirmed investigator-determined ORR per RECIST v1.1 DOR (by IRR and investigator) TTR (by IRR and investigator)[†] DCR (by IRR and investigator) PFS (by IRR and investigator) OS 		
• To evaluate the safety and tolerability of encorafenib plus binimetinib in treatment-naive and previously treated patients with <i>BRAF</i> ^{V600E} -mutant NSCLC	 Incidence and severity of AEs graded according to NCI CTCAE v4.03 Changes in clinical laboratory test parameters, vital signs, ECGs and echocardiogram/MUGA scans 		
Exploratory objectives	Exploratory end points		
\bullet To evaluate the PK of encorafenib and its metabolite LHY746 and binimetinib in patients with BRAF $^{\rm V600}\text{-}mutant$ NSCLC	 Plasma concentration-time profiles and PK parameter estimates for encorafenib and its metabolite LHY746 and binimetinib 		
• To assess blood ctDNA mutation status	Genomic analysis of ctDNA in blood samples		
[†] End point will be apalyzed but was not prespecified in the study protocol			

AE: Adverse event DCR: Disease control rate; DOR: Duration of response; ECG: Electrocardiogram; IRR: Independent radiology review; MUGA: Multigated acquisition; NCI CTCAE: National Cancer Institute Common Terminology Criteria for Adverse Events; NSCLC: Non-small-cell lung cancer; ORR: Objective response rate; OS: Overall survival; PFS: Progression-free survival; PK: Pharmacokinetics; RECIST: Response Evaluation Criteria in Solid Tumors; TTR: Time-to-tumor response.

Enrollment

PHAROS plans to enroll approximately 107 patients, including at least 60 treatment-naive and 37 previously treated patients with locally confirmed *BRAF*^{V600E} mutations. Up to ten additional patients with other *BRAF*^{V600} mutations may also be included. Patients are being enrolled at study sites in five countries: Italy (five sites), The Netherlands (two sites), South Korea (three sites), Spain (four sites) and the USA (39 sites).

Efficacy end points & assessments

Study objectives are summarized in Table 2. Antitumor activity as measured by ORR based on independent radiology review (IRR) is the primary end point. ORR is defined as the proportion of patients achieving a confirmed best overall response (complete response [CR] or partial response [PR]). ORR will be analyzed in treatment-naive and previously treated patients.

Secondary efficacy end points include: confirmed ORR as determined by the investigator; DOR (time from first confirmed CR or PR to earliest instance of disease progression or death), disease control rate at week 24 (DCR;

proportion of patients with a confirmed CR or PR or stable disease), PFS (time from first dose of study drug to earliest instance of disease progression or death) and time-to-tumor response (TTR; evaluated for responders as the time from first dose of study treatment to first confirmed objective response [CR or PR]), by IRR and investigator assessment; and OS (time from first dose of study treatment to death).

RECIST v1.1 criteria [39] will be used for tumor assessment and response. The imaging assessments are to be conducted at screening, every 8 weeks (56 ± 7 days) for 12 months, and every 12 weeks (± 7 days) thereafter until disease progression or the end of the study. Radiologic disease follow-up is to be continued in patients who permanently discontinue study treatment for a reason other than disease progression. Radiologic images from screening and follow-up are to be sent to a central imaging vendor for IRR. Per RECIST v1.1, responses (CR and PR) require confirmation by repeat radiographic assessments performed at least 4 weeks after the criteria for response are first met and no later than the next per-protocol scheduled scan, whichever is clinically indicated.

The OS follow-up is to be continued every 12 weeks (\pm 7 days) after the last dose of study treatment until withdrawal of consent, loss to follow-up, death or end of study. Subsequent anticancer therapies are also to be recorded during the survival follow-up period.

Safety end points & assessments

Safety and tolerability are evaluated as secondary objectives (Table 2). AEs and serious AEs (SAEs) are to be monitored throughout the treatment period and during the 30-day post-treatment follow-up. In patients starting new anticancer therapy within 30 days after the end of study treatment, safety follow-up is to be continued up to the start of the new therapy.

AEs of special interest for PHAROS were established based on the COLUMBUS trial [17], including left ventricular dysfunction, hemorrhage, venous thromboembolism, ocular toxicities, pneumonitis/interstitial lung disease, hepatotoxicity, creatine phosphokinase elevations or rhabdomyolysis, and QTc prolongation.

Physical examinations and evaluation of vital signs, Eastern Cooperative Oncology Group performance status and clinical laboratory test parameters (hematology/chemistries) are to be obtained at screening, on day 1 of each treatment cycle, at the end of study treatment, and at the safety follow-up visit 30 days after the end of study treatment. Dermatologic exams, electrocardiography, echocardiogram/multigated acquisition (MUGA) scans and visual acuity assessments are also to be performed to investigate potential class-specific AEs.

Pharmacokinetic & biomarker assessments

Prespecified exploratory end points will evaluate pharmacokinetic (PK) parameters for encorafenib (and its metabolite LHY746) and binimetinib, as well as blood–ctDNA mutation status. For patients enrolling before protocol amendment 4 (dated 16 February 2021), serial blood samples for plasma PK analysis of encorafenib, LHY746 (encorafenib metabolite) and binimetinib were to be collected at 0.5 (\pm 5 min), 1.5 (\pm 5 min), 3 (\pm 10 min) and 6 (\pm 20 min) h postdose on day 1 and day 15 of the first treatment cycle, and within 30 min predose on Day 1 of the second cycle. For patients enrolling after protocol amendment 4, predose PK samples are to be collected within 30 min prior to dosing on Day 1 of treatment cycles 1–6. On PK visit days, morning doses of study drug are administered under observation by site personnel, after collecting the predose PK sample.

For ctDNA biomarker assessments, blood samples are to be collected at screening, on day 1 of treatment cycle 2 and each treatment cycle thereafter, and at the end of study treatment. Purified ctDNA will be analyzed for potential genomic markers of encorafenib and/or binimetinib activity. Blood samples may also be used for additional exploratory research (e.g., investigating genetic variants in ctDNA, such as *BRAF*^{N600} mutations, and additional tumor mutations). Peripheral blood samples (serum) will also be used for analysis of potential proteomic or metabolomic factors and signals.

Statistical analysis methods

The primary efficacy and safety analyses will be conducted using the safety analysis set, which will include all patients who receive ≥ 1 dose of study treatment. A sensitivity analysis of the primary end point (ORR by IRR) will be conducted in the response evaluable set, comprising patients in the safety analysis set with an adequate baseline disease assessment who meet at least one of the following criteria: have ≥ 1 postbaseline disease assessment ≥ 6 weeks from first dose, or withdrew from the study or experienced progressive disease or death at any time during the study. Data will be summarized for treatment-naive patients, previously treated patients and overall.

For the primary efficacy end point, the study was designed to test the null hypothesis of ORR \leq 39% for treatmentnaive patients (n = 60) and \leq 20% for previously treated patients (n = 37) with *BRAF*^{V600E}-mutant NSCLC, which are considered not sufficiently clinically meaningful compared with available therapies. For treatment-naive patients, the null hypothesis was based on results observed in patients with NSCLC with a PD-L1 tumor proportion score of \geq 50% who received pembrolizumab as a single agent (KEYNOTE-042), in which the ORR per IRR was 39% (95% CI: 34–45) [40]. For previously treated patients, the null hypothesis was based on the ORR of 18% (95% CI: 14–23) observed in previously treated patients with NSCLC with a PD-L1 tumor proportion score of \geq 1% who received single-agent pembrolizumab [41].

The sample size calculation is based on the primary efficacy end point. With 60 evaluable treatment-naive patients with *BRAF* ^{V600E}-mutant NSCLC, the power is more than 95% to test the null hypothesis of ORR \leq 39% versus the alternative hypothesis assuming an alternative target rate of 65% with a one-sided $\alpha \leq$ 0.025 based on a single-stage design using the exact test. The null hypothesis will be rejected if \geq 32 confirmed objective responses are observed among the 60 patients. With 37 evaluable previously treated patients with *BRAF* ^{V600E}-mutant NSCLC, the study has at least 90% power to test the null hypothesis of ORR \leq 20%, assuming an alternative target rate of 45% with a one-sided $\alpha \leq$ 0.025 as described above. The null hypothesis will be rejected if \geq 13 objective responses are observed among the 37 previously treated patients.

The ORR and DCR will be calculated with corresponding exact two-sided Clopper–Pearson 95% CIs. For analyses of PFS, DOR (among patients achieving a confirmed response) and OS, estimates of survival functions will be constructed using the Kaplan–Meier (KM) method [42]; medians and 95% CIs will be provided. For PFS and OS, KM curves will be generated. TTR will be calculated for patients with a confirmed objective tumor response and summarized using descriptive statistics.

For analysis of safety, treatment-emergent AEs and SAEs will be summarized (counts and percentages) by MedDRA preferred term and system organ class. AEs are classified by severity, based on National Cancer Institute Common Terminology Criteria for Adverse Events Version 4.03, and relationship to study drug will be assigned by the investigator. AEs and SAEs of special interest (all-causality and treatment-related) will be summarized by frequency and severity. All deaths occurring during the treatment period, within 30 days after the last dose of study treatment and more than 30 days after the last dose of study treatment will be summarized.

Discussion & future perspective

There is an unmet clinical need for additional effective, less toxic, therapeutics for patients with *BRAF*-mutant metastatic NSCLC.

The PHAROS trial was initiated on 4 June 2019 and is ongoing at the time of this report. Data from PHAROS will provide insight into the antitumor activity and safety of encorafenib plus binimetinib in patients with *BRAF* ^{V600}-mutant NSCLC. While PHAROS represents the first clinical trial of encorafenib plus binimetinib in NSCLC, evidence supporting the clinical activity of this combination against *BRAF*-driven tumors along with a manageable side-effect profile comes from patients with unresectable or metastatic *BRAF* ^{V600}-mutant melanoma, where the combination is one of the MEK/BRAF inhibitor standard-of-care regimens [18,22,33,34]. The dose modifications for toxicity management for PHAROS are consistent with the recommendations for this approved combination in patients with *BRAF* ^{V600E/K}-mutant unresectable or metastatic melanoma [37,38].

PHAROS will also collect information about AEs of interest that are consistent with the known class-based safety profile of BRAF and MEK inhibitors [16,17]. Safety concerns associated with this class of drugs in combination include skin toxicities, gastrointestinal disorders, pyrexia, ocular events and musculoskeletal and cardiovascular events [16,17,19,22,24,25,33,34]. While some tolerability issues with this class of drugs in combination (e.g., most cutaneous and ophthalmologic side effects) are generally manageable and do not typically result in treatment discontinuation, others (e.g., cardiovascular events including left ventricular dysfunction), while less common, are associated with treatment discontinuation [16]. Some safety concerns have been identified as particularly relevant for individual BRAF and MEK inhibitors (e.g., photosensitivity is common with vemurafenib while pyrexia is an important side effect of dabrafenib; both are less common with encorafenib) [16]. Coadministration of encorafenib 450 mg q.d with binimetinib 45 mg b.i.d reduces the incidence of drug-related-specific toxicities for binimetinib (acneiform dermatitis, peripheral edema and rash) and encorafenib (arthralgia, myalgia, headache, rash, and palmoplantar keratoderma) [16,17,43]. Encorafenib plus binimetinib could represent a potential alternative BRAF/MEK inhibitor combination with a unique safety profile for the treatment of metastatic NSCLC with *BRAF* mutations, adding a new treatment option for this subset of patients.

Limitations of this study include the lack of a control arm. However, the small sample size in PHAROS is consistent with a molecularly targeted approach intended for tumors driven by relatively uncommon oncogenes [24,25,44]. The cohort of patients who received prior treatment in the metastatic setting is heterogenous in terms of the nature of the prior therapy (i.e., immunotherapy, immunotherapy plus chemotherapy, and chemotherapy alone), potentially complicating analysis in this cohort. Given the challenges of conducting an adequately powered randomized trial in a disease with the rarity of *BRAF*-mutant NSCLC, a nonrandomized, single-arm trial design was implemented. The primary end point in PHAROS (ORR by IRR) is intended to provide evidence of antitumor activity. Indeed, nonrandomized trials showing meaningful and durable ORR and DOR have been used to support regulatory approvals for other treatments targeting unique molecular drivers in NSCLC [24,44,45]. In this study, the primary end point is based on IRR to help overcome potential bias due to the lack of a control arm. While PHAROS allows for enrollment of a small number (≤ 10) of patients with less common *BRAF*^{V600} mutations, the study is not designed to draw statistical conclusions about response rates in patients with *BRAF*^{V600} mutations other than V600E. Given this is a small phase II study, a pragmatic prospective study with a larger number of patients would be needed to confirm the results of this study.

PHAROS will enroll a biomarker-selected patient population, based on the known mechanism of action of encorafenib. Molecular testing for *BRAF* mutations is consistent with the current standard of care during the clinical evaluation of advanced or metastatic NSCLC [28]. The study includes a relatively broad population in terms of prior treatments for metastatic disease. Importantly, the study includes both a treatment-naive cohort and a cohort of patients who have progressed on one prior line of treatment. Furthermore, given the recent approvals of PD-1/PD-L1 inhibitors and their progressive use in the first-line treatment setting, the PHAROS trial includes patients who received prior first-line immunotherapy. It should also be noted that there is an ongoing phase II trial of encorafenib with binimetinib in patients with *BRAF* ^{V600E}-mutant metastatic NSCLC, which is being conducted in France (Clinical Trials identifier: NCT04526782; study ID number: IFCT-1904).

Conclusion

Herein, we describe the study design of PHAROS, an ongoing, open-label, single-arm, phase II trial evaluating the BRAF inhibitor encorafenib plus the MEK inhibitor binimetinib in patients with metastatic *BRAF* ^{V600}-mutant NSCLC. PHAROS includes both treatment-naive patients and patients who have progressed on a prior line of immunotherapy and/or chemotherapy. Results from the PHAROS trial will provide insight into the antitumor activity and safety of encorafenib plus binimetinib in patients with *BRAF* ^{V600}-mutant metastatic NSCLC.

Executive summary

Background

- BRAF and MEK inhibitor combinations can achieve greater antitumor activity than each as a single agent and are
 associated with a more favorable side effect profile compared with BRAF inhibitor monotherapy.
- BRAF and MEK inhibitor combinations have been validated clinically in *BRAF* ^{V600E/K}-mutant metastatic melanoma, where they are an established standard of care.
- BRAF oncogenic driver mutations occur in a subset of 3–4% of patients with non-small-cell lung cancer (NSCLC), and the BRAF inhibitor dabrafenib plus the MEK inhibitor trametinib demonstrated efficacy and safety in BRAF V^{600E}-mutant metastatic NSCLC.
- Encorafenib and binimetinib are selective and potent BRAF and MEK inhibitors, respectively, that have demonstrated efficacy and a distinct, favorable safety profile compared with other anti-BRAF/MEK-targeted therapies in patients with *BRAF* ^{V600E/K}-mutant metastatic melanoma.

PHAROS trial

- The PHAROS trial was initiated to explore the antitumor activity and safety of encorafenib plus binimetinib in
 patients with BRAF V600-mutant metastatic NSCLC who are either treatment-naive or who have been previously
 treated with platinum-based chemotherapy and/or anti-PD-1/PD-L1 inhibitor therapy.
- Approximately 107 patients, including at least 60 treatment-naive and 37 previously treated patients with BRAF V^{600E}-mutant metastatic NSCLC, will receive encorafenib 450 mg once daily plus binimetinib 45 mg twice daily in continuous 28-day cycles until disease progression, unacceptable toxicity, withdrawal of consent, initiation of subsequent anticancer therapy, death or end of study.
- The primary end point is the objective response rate, based on independent radiology review.
- Information about adverse events will be collected, including adverse events of interest, based on the known safety profile of this class of drugs in combination.
- Results from PHAROS will provide insight into the antitumor activity and safety of encorafenib plus binimetinib in patients with BRAF ^{V600}-mutant metastatic NSCLC.

Supplementary data

An infographic accompanies this paper. To view or download this infographic in your browser please click here: http://www.futuremedicine.com/doi/suppl/10.2217/fon-2021-1250

Author contributions

All the authors met the criteria for authorship set forth by the International Committee of Medical Journal Editors, and were involved in the conception, preparation and approval of the manuscript.

Acknowledgments

The authors thank the study participants.

Financial & competing interests disclosure

This study was sponsored by Array Biopharma, which was acquired by Pfizer in July 2019. GJ Riely is an investigator conducting contracted research for Merck, Mirati, Novartis, Pfizer, Roche and Takeda; received editorial support from Merck, Mirati, Novartis, Pfizer, Roche and Takeda. M-J Ahn has participated in advisory boards for Alpha Pharmaceuticals, Amgen, AstraZeneca, Bristol Myers Squibb, Eli Lilly, Merck, Merck Sharp & Dohme, ONO, Pfizer, Roche and Takeda. E Felip has participated in advisory boards for Amgen, AstraZeneca, Bayer, Beigene, Boehringer Ingelheim, Bristol Myers Squibb, Eli Lilly, GlaxoSmithKline, Janssen, Medical Trends, Merck Serono, Merck Sharp & Dohme, Peptomyc, Pfizer, Puma, Regeneron, Roche, Sanofi, Syneos Health and Takeda; has served on a speaker's bureau for Amgen, AstraZeneca, Bristol Myers Squibb, Eli Lilly, Janssen, Medscape, Merck Serono, Merck Sharp & Dohme, Peervoice, Pfizer, Roche, Springer and Touch Medical. SS Ramalingam has served as a consultant for Amgen, AstraZeneca, Bristol Myers Squibb, Daiichi Sankyo, Eisai, Eli Lilly, Genentech/Roche, GlaxoSmithKline, Merck, Sanofi and Takeda; has received research grants from Advaxis, Amgen, AstraZeneca, Bristol Myers Squibb, Genmab, GlaxoSmithKline, Merck, Takeda and Tesaro. EF Smit has served as an advisor/consultant for AstraZeneca, Bayer, Bristol Myers Squibb, Cellgene, Daiichi Sankyo, Eli Lilly, Genentech/Roche, Merck, Merck Sharp & Dohme, Novartis, Pfizer, Regeneron, Roche Diagnostics and Takeda; has received research grants from AstraZeneca, Bristol Myers Squibb, Genentech/Roche, Merck, and Merck Sharp & Dohme. AS Tsao has received research grants from Boehringer Ingelheim, Eli Lilly, EMD Serono, Epizyme, Genentech, Millennium, Polaris, Seattle Genetics and Takeda; participated in advisory boards for Ariad, AstraZeneca, Boehringer Ingelheim, Bristol Myers Squibb, Genentech/Roche, Eli Lilly, EMD Serono, Merck, Novartis, Seattle Genetics, Sellas Life Science and Takeda. A Alcasid, T Usari, PS Wissel and KD Wilner are employed by and own stock in Pfizer. BE Johnson has received research support from Cannon Medical Systems and Novartis; has served as a consultant/advisor for Hengrui Therapeutics and Novartis; participated in advisory boards for Boston Pharmaceuticals, Checkpoint Therapeutics, Daiichi Sankyo, G1 Therapeutics, Genentech, Janssen Scientific and Jazz Pharmaceuticals; received post-marketing royalties for EGFR testing from the Dana-Farber Cancer Institute. The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

Medical writing support was provided by Erin Bekes on behalf of CMC AFFINITY, McCann Health Medical Communications, and was funded by Pfizer.

Ethical conduct of research

The study is being conducted in accordance with the protocol, legal and regulatory requirements, and the general principles set forth in the International Ethical Guidelines for Biomedical Research Involving Human Subjects (Council for International Organizations of Medical Sciences 2002), International Council for Harmonisation Guideline for Good Clinical Practice and the Declaration of Helsinki.

Data-sharing statement

Upon request, and subject to review, Pfizer will provide the data that support the findings of this study. Subject to certain criteria, conditions and exceptions, Pfizer may also provide access to the related individual deidentified participant data. See https://www.pfizer.com/science/clinical-trials/trial-data-and-results for more information.

Open access

This work is licensed under the Attribution-NonCommercial-NoDerivatives 4.0 Unported License. To view a copy of this license, visit http://creativecommons.org/licenses/by-nc-nd/4.0/

References

Papers of special note have been highlighted as: • of interest

- Planchard D, Popat S, Kerr K et al. Metastatic non-small cell lung cancer: ESMO Clinical Practice Guidelines for diagnosis, treatment and follow-up. Ann. Oncol. 29(Suppl. 4), iv192–iv237 (2018).
- 2. König D, Savic Prince S, Rothschild SI. Targeted therapy in advanced and metastatic non-small cell lung cancer. an update on treatment of the most important actionable oncogenic driver alterations. *Cancers* 13(4), 804 (2021).
- 3. Howlader N, Forjaz G, Mooradian MJ *et al.* The effect of advances in lung-cancer treatment on population mortality. *N. Engl. J. Med.* 383(7), 640–649 (2020).
- Barlesi F, Mazieres J, Merlio J-P *et al.* Routine molecular profiling of patients with advanced non-small-cell lung cancer: results of a 1-year nationwide programme of the French Cooperative Thoracic Intergroup (IFCT). *Lancet* 387(10026), 1415–1426 (2016).
- Cardarella S, Ogino A, Nishino M et al. Clinical, pathologic, and biologic features associated with BRAF mutations in non-small cell lung cancer. Clin. Cancer Res. 19(16), 4532–4540 (2013).
- Lin L, Asthana S, Chan E et al. Mapping the molecular determinants of BRAF oncogene dependence in human lung cancer. Proc. Natl Acad. Sci. USA 111(7), E748–E757 (2014).
- 7. Alvarez JGB, Otterson GA. Agents to treat BRAF-mutant lung cancer. Drugs Context 8, 212566 (2019).
- 8. Frisone D, Friedlaender A, Malapelle U, Banna G, Addeo A. A BRAF new world. Crit. Rev. Oncol. Hematol. 152, 103008 (2020).
- 9. O'Leary CG, Andelkovic V, Ladwa R et al. Targeting BRAF mutations in non-small cell lung cancer. Transl. Lung Cancer Res. 8(6), 1119–1124 (2019).
- 10. Roviello G, D'angelo A, Sirico M, Pittacolo M, Conter FU, Sobhani N. Advances in anti-BRAF therapies for lung cancer. *Invest. New Drugs* 39(3), 879–890 (2021).
- 11. Baik CS, Myall NJ, Wakelee HA. Targeting *BRAF*-mutant non-small cell lung cancer: from molecular profiling to rationally designed therapy. *Oncologist* 22(7), 786–796 (2017).
- 12. Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. *CA Cancer J. Clin.* 68(6), 394–424 (2018).
- 13. Litvak AM, Paik PK, Woo KM *et al.* Clinical characteristics and course of 63 patients with *BRAF* mutant lung cancers. *J. Thorac. Oncol.* 9(11), 1669–1674 (2014).
- Flaherty KT, Infante JR, Daud A et al. Combined BRAF and MEK inhibition in melanoma with BRAF V600 mutations. N. Engl. J. Med. 367(18), 1694–1703 (2012).
- 15. Menzer C, Menzies AM, Carlino MS et al. Targeted therapy in advanced melanoma with rare *BRAF* mutations. J. Clin. Oncol. 37(33), 3142–3151 (2019).
- Heinzerling L, Eigentler TK, Fluck M et al. Tolerability of BRAF/MEK inhibitor combinations: adverse event evaluation and management. ESMO Open 4(3), e000491 (2019).
- A comprehensive review of the management of the clinically relevant side effects of three combinations of BRAF and MEK inhibitors in patients with *BRAF* ^{V600}-mutant melanoma.
- 17. Gogas HJ, Flaherty KT, Dummer R *et al.* Adverse events associated with encorafenib plus binimetinib in the COLUMBUS study: incidence, course and management. *Eur. J. Cancer* 119, 97–106 (2019).
- A detailed analysis of adverse events in the phase 3 COLUMBUS trial, which evaluated the combination of encorafenib plus binimetinib in patients with *BRAF*^{V600}-mutant melanoma.
- National Comprehensive Cancer Network[®]. NCCN Clinical Practice Guidelines in Oncology (NCCN Guidelines). *Melanoma: Cutaneous Version 2.2021* (2021). www.nccn.org/professionals/physician_gls/pdf/cutaneous.melanoma.pdf
- Long GV, Stroyakovskiy D, Gogas H et al. Combined BRAF and MEK inhibition versus BRAF inhibition alone in melanoma. N. Engl. J. Med. 371(20), 1877–1888 (2014).
- Ascierto PA, Mcarthur GA, Dréno B et al. Cobimetinib combined with vemurafenib in advanced BRAF ^{V600}-mutant melanoma (coBRIM): updated efficacy results from a randomised, double-blind, phase 3 trial. Lancet Oncol. 17(9), 1248–1260 (2016).
- Robert C, Karaszewska B, Schachter J et al. Improved overall survival in melanoma with combined dabrafenib and trametinib. N. Engl. J. Med. 372(1), 30–39 (2015).
- 22. Dummer R, Ascierto PA, Gogas HJ *et al.* Encorafenib plus binimetinib versus vemurafenib or encorafenib in patients with *BRAF*-mutant melanoma (COLUMBUS): a multicentre, open-label, randomised phase 3 trial. *Lancet Oncol.* 19(5), 603–615 (2018).
- Primary paper reporting the results of part 1 of the phase 3 COLUMBUS trial which evaluated the combination of encorafenib plus binimetinib in patients with *BRAF*^{V600}-mutant melanoma.
- 23. Robert C, Grob JJ, Stroyakovskiy D *et al.* Five-year outcomes with dabrafenib plus trametinib in metastatic melanoma. *N. Engl. J. Med.* 381(7), 626–636 (2019).
- 24. Planchard D, Besse B, Groen HJM *et al.* Dabrafenib plus trametinib in patients with previously treated *BRAF* ^{V600E}-mutant metastatic non-small cell lung cancer: an open-label, multicentre phase 2 trial. *Lancet Oncol.* 17(7), 984–993 (2016).

- Phase 2 activity and safety results for dabrafenib plus trametinib in patients with previously treated BRAF V600E-mutant non-small-cell lung cancers (NSCLCs).
- Planchard D, Smit EF, Groen HJM *et al.* Dabrafenib plus trametinib in patients with previously untreated *BRAF* ^{V600E}-mutant metastatic non-small-cell lung cancer: an open-label, phase 2 trial. *Lancet Oncol.* 18(10), 1307–1316 (2017).
- Phase 2 activity and safety results for dabrafenib plus trametinib in patients with previously untreated BRAF V600E-mutant NSCLC.
- Planchard D, Kim TM, Mazieres J et al. Dabrafenib in patients with BRAF V^{600E}-positive advanced non-small-cell lung cancer: a single-arm, multicentre, open-label, phase 2 trial. Lancet Oncol. 17(5), 642–650 (2016).
- Planchard D, Besse B, Groen HJM *et al.* Phase 2 study of dabrafenib plus trametinib in patients with *BRAF* V600E-mutant metastatic non-small cell lung cancer: updated 5-year survival rates and genomic analysis. *J. Thorac. Oncol.* doi:10.1016/j.jtho.2021.08.011 (2021) (Epub ahead of print).
- Long-term phase 2 activity and safety results for dabrafenib plus trametinib in treatment-naive and previously treated patients with BRAFV600E-mutant NSCLC.
- 28. National Comprehensive Cancer Network[®]. NCCN Clinical Practice Guidelines in Oncology (NCCN Guidelines). *Non-Small Cell Lung Cancer Version 5.2021* (2021).www.nccn.org/professionals/physician_gls/pdf/nscl.pdf
- 29. Stuart D, Li N, Poon DJ et al. Preclinical profile of LGX818: a potent and selective RAF kinase inhibitor. Presented at: AACR 103rd Annual Meeting, Chicago, IL, USA 31 March 4 April 2012.
- 30. Lee PA, Wallace E, Marlow A *et al.* Preclinical development of ARRY-162, a potent and selective MEK 1/2 inhibitor. *Presented at: AACR 101st Annual Meeting.* Washington, DC, USA 17-21 April 2010.
- 31. Koelblinger P, Thuerigen O, Dummer R. Development of encorafenib for *BRAF*-mutated advanced melanoma. *Curr. Opin. Oncol.* 30(2), 125–133 (2018).
- 32. Delord JP, Robert C, Nyakas M *et al.* Phase I dose-escalation and -expansion study of the BRAF inhibitor encorafenib (LGX818) in metastatic *BRAF*-mutant melanoma. *Clin. Cancer Res.* 23(18), 5339–5348 (2017).
- Dummer R, Ascierto PA, Gogas HJ *et al.* Overall survival in patients with *BRAF*-mutant melanoma receiving encorafenib plus binimetinib versus vemurafenib or encorafenib (COLUMBUS): a multicentre, open-label, randomised, phase 3 trial. *Lancet Oncol.* 19(10), 1315–1327 (2018).
- Overall survival results from the phase 3 COLUMBUS trial which evaluated the combination of encorafenib plus binimetinib in patients with *BRAF* ^{V600}-mutant melanoma.
- 34. Flaherty K. In the pipeline: encorafenib and binimetinib in *BRAF*-mutated melanoma. *Clin. Adv. Hematol. Oncol.* 15(10), 745–747 (2017).
- 35. Ascierto PA, Robert C, Nathan PD *et al.* Pyrexia-related outcomes upon application of an adapted pyrexia management algorithm in patients (pts) with *BRAF* V600: Mutant unresectable or metastatic melanoma treated with dabrafenib plus trametinib (DabTram) in the COMBI-i trial. *J. Clin. Oncol.* 39(Suppl. 15), Abstract 9560 (2021).
- 36. Atkinson V, Robert C, Grob J-J *et al.* Improved pyrexia-related outcomes associated with an adapted pyrexia adverse event (AE) management algorithm in patients (pts) treated with adjuvant dabrafenib + trametinib (dab + tram): primary results of COMBI-APlus. *J. Clin. Oncol.* 39(Suppl. 15), Abstract 9525 (2021).
- 37. Pfizer Inc. BRAFTOVI[®] (encorafenib): Prescribing Information (2021). http://labeling.pfizer.com/ShowLabeling.aspx?id=12990#S2.1
- 38. Pfizer Inc. MEKTOVI[®] (binimetinib): Prescribing Information (2021). http://labeling.pfizer.com/ShowLabeling.aspx?id=12988
- 39. Eisenhauer EA, Therasse P, Bogaerts J *et al.* New response evaluation criteria in solid tumours: revised RECIST guideline (version 1.1). *Eur. J. Cancer* 45(2), 228–247 (2009).
- Mok TSK, Wu YL, Kudaba I *et al.* Pembrolizumab versus chemotherapy for previously untreated, PD-L1-expressing, locally advanced or metastatic non-small-cell lung cancer (KEYNOTE-042): a randomised, open-label, controlled, phase 3 trial. *Lancet* 393(10183), 1819–1830 (2019).
- 41. Herbst RS, Baas P, Kim DW *et al.* Pembrolizumab versus docetaxel for previously treated, PD-L1-positive, advanced non-small-cell lung cancer (KEYNOTE-010): a randomised controlled trial. *Lancet* 387(10027), 1540–1550 (2016).
- 42. Kaplan EL, Meier P. Nonparametric estimation from incomplete observations. J. Am. Stat. Assoc. 53(282), 457-481 (1958).
- 43. Koelblinger P, Dornbierer J, Dummer R. A review of binimetinib for the treatment of mutant cutaneous melanoma. *Future Oncol.* 13(20), 1755–1766 (2017).
- 44. Shaw AT, Ou SH, Bang YJ et al. Crizotinib in ROS1-rearranged non-small-cell lung cancer. N. Engl. J. Med. 371(21), 1963–1971 (2014).
- 45. Blumenthal GM, Pazdur R. Response rate as an approval end point in oncology: back to the future. JAMA Oncol. 2(6), 780-781 (2016).